Author's response to reviews

Title: Genetic mutations in SPINK1, CFTR, CTRC genes in acute pancreatitis

Authors:

Dorota Koziel (dorota.koziel@wp.pl)
Stanislaw Gluszek (sgluszek@wp.pl)
Artur Kowalik (artur_kowalik@yahoo.com)
Malgorzata Chlopek (malgo.chlopek@gmail.com)
Liliana Pieciak (lilianamazurek@o2.pl)

Version: 15
Date: 15 June 2015

Author's response to reviews:

Kielce, 16 June 2015

Dear Editor,

The following changes were introduced according to the reviewers’ comments:

Copyediting:

We recommend that you copyedit the paper to improve the style of written English.

The language has been revised from the editorial aspect by Edanz Group.

All authors have approved the manuscript and agree with its submission to BMC Gastroenterology.

Yours Faithfully,

Dorota Koziel, Stanisław Gluszek, Artur Kowalik, Malgorzata Chlopek, Liliana Pieciak